Author’s response to reviews

Title: Unplanned readmission rates, length of hospital stay, mortality, and medical costs of ten common medical conditions: a retrospective analysis of Hong Kong hospital data

Authors:

Eliza LY Wong (lywong@cuhk.edu.hk)
Annie WL Cheung (anniewlcheung@cuhk.edu.hk)
Michael CM Leung (cmleung123@hotmail.com)
Carrie HK Yam (carrieyam@cuhk.edu.hk)
Frank WK Chan (cwkfrank@cuhk.edu.hk)
Fiona YY Wong (fionawong@cuhk.edu.hk)
Eng Kiong Yeoh (yeoh_ek@cuhk.edu.hk)

Version: 6 Date: 14 June 2011

Author’s response to reviews:

1) Page 11: Changed the place of reference [42]
2) Page 12: Added "." after reference [48,49]